Author's response to reviews

Title: Mannose-Binding Lectin deficiency with eosinophilic meningoencephalitis due to Angiostrongylus cantonensis in children: a case series

Authors:

Barbara Padilla-Docal (barbara.padilla@infomed.sld.cu)
Alberto Juan Dorta-Contreras (adorta@infomed.sld.cu)
Raisa Bu-Coifiu-Fanego (raisabu@infomed.sld.cu)
Rene H Martinez-Alderete (urmalderete@infomed.sld.cu)
Olga Susana De Paula-Almeida (mariafdocal@infomed.sld.cu)
Hansotto Reiber (ho@horeiber.de)
Jens Christian Jensenius (j.c.jensenius@immunology.au.dk)

Version: 3 Date: 4 December 2010

Author's response to reviews:

The JMCR Editorial Team

Dear Sirs:

I should be glad to answer the comments and the suggestion of the Associate Editor about our paper MS: 1667095943431576 titled: Mannose-Binding Lectin deficiency with eosinophilic meningoencephalitis due to Angiostrongylus cantonensis in children: a case series

Barbara Padilla-Docal, Alberto Juan Dorta-Contreras, Raisa Bu-Coifiu-Fanego, Rene H Martinez-Alderete, Olga Susana De Paula-Almeida, Hansotto Reiber and Jens Christian Jensenius

Referee 1:

Table 1 and 2 must show the normal range values.

Answer:

We have already added it as you may see in the new version. The normal values are shown in table 1 and 2 in blue colour.

Referee 2:

This reviewer said: It would be interesting to increase the number of patients and a range of age to be studied in order to know the relationship among this parameters and MBL deficiency.

Answer: Of course this increment of the number of patients would be interesting but the frequency is very low. For this purpose we have decided to publish it as a case series report.
The suggestion of the Associate Editor.

Suggestion:
Tab 1 and 2 must show the normal range values.

Answer:
We have included the normal values in Table 1 and 2. Notice it in blue colour.

Suggestion:
Introduction; Page 2 second paragraph, miss a hyphen between H-Ficolin. also, missing space between pathogen & associated.

Answer
We put the hyphen in H-ficolin and put the space between pathogen and associated. The corrected words are in blue colour.

We put the comments in a revised manuscript and the present cover letter gives a point-by-point response to the concerns.

Many thanks and best regards,
Dr. Alberto J Dorta-Contreras
Corresponding author